Dear Editor,

Please find enclosed the new revised manuscript entitled “Re: Association between intestinal neoplasms and celiac disease: beyond celiac disease and more.”. I have read all the reviewers’ comments along with the most recent reviewer’s comments. I do appreciate the reviewers’ efforts in pointing out the areas for improvement.

I have made changes to the manuscript taking into consideration the comments made by the reviewers. In particular, I have added other findings and changed the way to suggest regarding the pathogenesis of enteropathy associated T cell lymphoma. All major changes in the new manuscript are highlighted in yellow in the new manuscript. I hope the changes are sufficient with respect to the comments raised by the reviewers. I look forward to your positive response.

With my best regards,

Sincerely,

Kenji Okumura
Reply to reviewers:

Reviewer #1:

Specific Comments to Authors: I have nothing against Editorials, which usually directly highlight to some interesting part of selected manuscript, or they critically comment results, or conclusions, but this is only short general comment. Now everybody can write that they agree with results, or they principally like results, or conclusions. So finally I just would like to suggest to authors to write short review to this topic, or short communication and here compare their results with this paper, including another important articles dealing with malignancy and celiac disease. That article can be accepted for publication and can be also attractive for readers.

Answer:

I like the thank the reviewer for a thorough review of this manuscript.

I would like to highlight some of the findings, which were not discussed in the original article.

With the current revision, I have expanded more findings from other authors, which are associated with the pathogenesis of enteropathy associated T cell lymphoma. I believe this will be helpful for the readers in this journal.

Reviewer #2:

Specific Comments to Authors: It is interesting letter but need more details.

Answer:

I like to thank the reviewer on this comments. I have reviewed my previous manuscript and I have added more findings based on the previous published paper.